Author's response to reviews

Title: Prospective study of asbestos-related diseases incidence cases in primary health care in an area of Barcelona province

Authors:

Magdalena Rosell Murphy (mrosell@idiapigol.org)
Rafael Abós- Herràndiz (24025rah@comb.cat)
Josep Tarrés (7893ito@comb.cat)
Xavier Martínez-Artés (xmartinezartes@yahoo.es)
Isabel García-Allas (26584iga@comb.cat)
Ilona Krier (ilonakrier@yahoo.es)
Gloria Cantarell (gcantarell@tauli.cat)
Miguel Gallego (mgallego@tauli.cat)
Ramon Orriols (rorriols@vhebron.net)
Constança Albertí (Constanca.alberti@gencat.cat)

Version: 2 Date: 11 March 2010

Author's response to reviews: see over
Cover letter

March, 11, 2010-03-11
Changes performed in the manuscript: MS 1224386532354790.

1. Abstract changed and structured in separate sections:
   Background
   Methods/ Design
   Discussion
   All abbreviations have been avoided in this section

2. Power calculation: This is a population based study, so we are aiming to include all cases diagnosed in the area. This issue is stated in the abstract and also in the methods section (page 17)

3. Statement with the name of the ethical committee that has approved the study (page 19)

4. I have changed the order of authors in the document attached, to comply with the order of authors included already in the biomedical web sent in the first draft.

I have highlighted all changes in the text I am including.